Author's response to reviews

Title: A randomised controlled trial to prevent hospital readmissions and loss of functional ability in high risk older adults: a study protocol

Authors:

Mary D Courtney (mary.courtney@ubc.ca)
Helen E Edwards (h.edwards@qut.edu.au)
Anne M Chang (am.chang@qut.edu.au)
Anthony W Parker (t.parker@qut.edu.au)
Kathleen Finlayson (k.finlayson@qut.edu.au)
Kyra Hamilton (k2.hamilton@qut.edu.au)

Version: 5 Date: 22 August 2011

Author's response to reviews:

In the authors' contributions section we now indicate that all authors have read and approved the final manuscript.